Author's response to reviews

Title: Age at onset of Huntington disease is not modulated by the R72P variation in TP53 and the R196K variation in the gene coding for the human caspase activated DNase (hCAD)

Authors:

Larissa Arning (larissa.arning@rub.de)
Peter H Kraus (peter.h.kraus@rub.de)
Carsten Saft (carsten.saft@cityweb.de)
Jurgen Andrich (juergen.andrich@rub.de)
Joerg T Epplen (joerg.t.epplen@rub.de)

Version: 4 Date: 22 September 2005

Author's response to reviews:

Dear Editor:

Herewith we resubmit the revised version of our manuscript:

Title & Abstract - Please ensure these are identical in the manuscript and in the submission system.

We checked this for correctness.

Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system.

The authors are written identically, beside the umlauts.

Competing interests - If there are none to declare, please write only 'The authors declare that they have no competing interests'.

We substituted this sentence.

Thank you and your referees for reviewing the manuscript.

Sincerely

Larissa Arning